Author's response to reviews

Title: The effectiveness of pelvic floor muscle training alone and with other physical therapies for the treatment of stress urinary incontinence in women: a systematic review

Authors:
  Patricia B Neumann (cpneumann@ozemail.com.au)
  Karen A Grimmer (Karen.grimmer@unisa.edu.au)
  Yamini Deenadayalan (Yamini.deenadayalan@unisa.edu.au)

Version: 7  Date: 18 May 2006

Author's response to reviews: see over
Cover Letter with details of changes

Page 1, line 1

In order to better reflect the intention of the review and to conform with the terminology of the main text, the title has been changed from:

**Pelvic floor muscle training and other physical therapies for the treatment of stress urinary incontinence in women: a systematic review**

to:

**Pelvic floor muscle training and adjunctive therapies for the treatment of stress urinary incontinence in women: a systematic review**

There have been no other changes other than those requested and minor formatting changes.